

RESEARCH ARTICLE

10.1029/2017WR022478

Special Section:

Hydrology delivers Earth System Sciences to Society (HESSS4): Improving and Integrating Knowledge across Disciplines on Global Energy, Water and Carbon Cycles

Key Points:

- A new large-scale hydroeconomic model (ECHO) is developed
- ECHO is used to evaluate the water management adaptation pathways in Africa under various socioeconomic and climate futures
- Future scenario simulations highlight the capacity of ECHO to address challenging research questions related to water-energy-land nexus

Supporting Information:

Supporting Information S1

Correspondence to: T. Kahil, kahil@iiasa.ac.at

Citation:

Kahil, T., Parkinson, S., Satoh, Y., Greve, P., Burek, P., Veldkamp, T. I. E., et al. (2018). A continental-scale hydroeconomic model for integrating water-energy-land nexus solutions. *Water Resources Research*, 54. https:// doi.org/10.1029/2017WR022478

Received 28 DEC 2017 Accepted 8 SEP 2018 Accepted article online 19 SEP 2018

©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

A Continental-Scale Hydroeconomic Model for Integrating Water-Energy-Land Nexus Solutions

Taher Kahil¹^(D), Simon Parkinson^{1,2}, Yusuke Satoh¹^(D), Peter Greve¹^(D), Peter Burek¹^(D), Ted I. E. Veldkamp^{1,3}, Robert Burtscher¹, Edward Byers¹^(D), Ned Djilali²^(D), Guenther Fischer¹, Volker Krey¹, Simon Langan¹, Keywan Riahi¹, Sylvia Tramberend¹^(D), and Yoshihide Wada^{1,4}^(D)

¹International Institute for Applied Systems Analysis, Laxenburg, Austria, ²Institute for Integrated Energy Systems, University of Victoria, Victoria, British Columbia, Canada, ³Institute for Environmental Studies (IVM), Vrije Universiteit Amsterdam, Amsterdam, Netherlands, ⁴Department of Physical Geography, Utrecht University, Utrecht, Netherlands

Abstract This study presents the development of a new bottom-up large-scale hydroeconomic model, Extended Continental-scale Hydroeconomic Optimization (ECHO), that works at a subbasin scale over a continent. The strength of ECHO stems from the integration of a detailed representation of local hydrological and technological constraints with regional and global policies, while accounting for the feedbacks between water, energy, and agricultural sectors. In this study, ECHO has been applied over Africa as a case study with the aim of demonstrating the benefits of this integrated hydroeconomic modeling framework. Results of this framework are overall consistent with previous findings evaluating the cost of water supply and adaptation to global changes in Africa. Moreover, results provide critical assessments of future investment needs in both supply- and demand-side water management options, economic implications of contrasting future socioeconomic and climate change scenarios, and the potential trade-offs among economic and environmental objectives. Overall, this study demonstrates the capacity of ECHO to address challenging research questions examining the sustainability of water supply and the impacts of water management on energy and food sectors and vice versa. As such, we propose ECHO as useful tool for water-related scenario analysis and management options evaluation.

1. Introduction

Global water withdrawals have been increasing rapidly during the last decades in order to sustain growing food and energy demands and increasing standards of living (Kummu et al., 2010; Liu et al., 2017; Mekonnen & Hoekstra, 2016). As a result, many basins around the world have experienced pervasive water scarcity conditions and related water management challenges (Kahil, Dinar, & Albiac, 2015; Veldkamp et al., 2017; Wada et al., 2013). These challenges are expected to become more critical in the coming decades as countries attempt to sustain a larger and more prosperous human population and economy under changing climate conditions (Hanasaki et al., 2013; Kim et al., 2016). As such, policymakers in vulnerable basins need to anticipate on how to adapt management practices to secure reliable future water supply that can meet the demands from different sectors. However, the choice of water management options is often associated with trade-offs across multiple water-related systems such as food production, energy supply, and ecosystem services, as well as across space and time (Banzhaf, 2009; Hurford et al., 2014). An appropriate choice of these options calls for the development of a systematic approach, depicting the biophysical and socioeconomic factors that determine the future dynamics of river basins, including the key interactions among water, energy, and agricultural systems (Brown et al., 2015; Rogers & Fiering, 1986; Wada et al., 2017).

In recent decades, hydroeconomic (HE) models have emerged as an important tool for informing basin-scale water resources planning because they include an integrated biophysical-technological-economic representation of the water resources systems (Bekchanov et al., 2017; Harou et al., 2009). These features are usually represented using a set of physical and technology choice equations (or core model). Numerical optimization algorithms are then applied to the core model to back calculate a set of primary decisions that collectively result in the best feasible outcome from the perspective of specific objectives important to decision-making (Booker et al., 2012). For example, an economic objective that focuses on minimizing costs or maximizing benefits is typical in HE models because it facilitates valuation of resource and policy constraints (Ward, 2009). Similarly, simulation algorithms can be used in HE models to more

<mark>,</mark>

realistically represent complex water systems with nonlinear physical or institutional processes. Traditionally, HE models have been used to evaluate the efficiency of alternative water allocation mechanisms under existing infrastructure (Booker et al., 2005; Booker & Young, 1994; Cai et al., 2003; Kahil, Dinar, & Albiac, 2015; Noel & Howitt, 1982; Ward & Lynch, 1997) and to identify bottlenecks in the water system, where investments in new infrastructure would be most beneficial (Acquah & Ward, 2017; Gohar et al., 2013; Qureshi et al., 2010). Recently, HE models have also been used to assess how effectively the water system can adapt to future climatic and socioeconomic changes and explore the value of various options for doing this (Connor et al., 2009; Escriva-Bou et al., 2017; Kahil et al., 2016; Medellín-Azuara et al., 2008; Tanaka et al., 2006).

In the literature, HE modeling is rarely used across large spatial scales. More typically, HE models have been designed at basin scales (Harou et al., 2009), with a few designed to model systems ranging from household or utility scales (Alcubilla & Lund, 2006; Jenkins & Lund, 2000; Rosenberg et al., 2007) to transboundary basin scales (Fisher et al., 2002; Nigatu & Dinar, 2016; Ringler et al., 2004). The few large spatial scale models include national scale examples, such as the statewide HE model of California (CALVIN; Draper et al., 2003), and global-scale examples, such as the IMPACT-WATER (Cai & Rosegrant, 2002) and IGSM-WRS (Strzepek et al., 2013) models. Advantageously, these large-scale HE models provide the opportunity to integrate a detailed representation of local biophysical and technological constraints with farther-reaching regional and global policies. This feature is particularly relevant because the availability of water, energy, and land resources varies significantly at local scales, whereas the linkage to regional and international markets for energy and food commodities and transboundary treaties for water resources results in global influences (Dalin et al., 2017; Holland et al., 2015; Zawahri et al., 2016). However, existing large-scale HE models use a reduced number of spatial units in order to minimize the computational burden, which limits their potential for integrating properly local-level constraints. Moreover, these large-scale HE models have been designed to identify bottlenecks in the water system, but they have not been designed to identify the least cost optimal combination of water management options for meeting growing and changing water demands. Likewise, many omit the implications of management options in the energy and agricultural sectors such as investments into thermal power plant cooling technologies, irrigation systems, and trade opportunities.

To meet this specific need and consistently address water-energy-land nexus management solutions, we develop the Extended Continental-scale Hydroeconomic Optimization (ECHO) model: a new large-scale HE model for long-term water management planning. ECHO is an extended HE model because it includes both capacity expansion and operational decisions for a wide range of water management options in the water system, as well as in energy and agricultural systems, with the choice among these options is always economically evaluated to reveal opportunities for efficiency gains. This feature is particularly relevant, given that adaptation to future water scarcity is a challenging task, especially for regions with limited economic capacity to invest in capital-intensive water infrastructure and management options. In these regions, a better alignment of policies across water-related domains and regions is needed in order to reduce overall investment costs. Moreover, ECHO covers an extensive number of subbasin units within a reduced-form transboundary river network. This feature enables identifying the spatial differences in supply and demand patterns and the choice of adaptation pathways. Lastly, the ability of ECHO to combine various components, including hydrology, agriculture and energy uses, and economics into a holistic large-scale modeling framework, which is solved in its entirety where information between components is transferred endogenously, facilitates a more effective integrated optimization of water-energy-land nexus management solutions. Therefore, ECHO could identify a broader solution space, achieving overall efficiency of water-energy-land resources utilization and producing synergistic benefits across large spatial domains (Cai et al., 2018). In this paper, we apply ECHO to Africa as a case study in order to assess important interactions between the region's future water demand and availability under various socioeconomic and climatic scenarios. However, ECHO is currently designed to operate at different spatial scales and in different regions or continents, given the data availability and its application to Africa aims only to highlight the benefits of ECHO model development, rather than to recommend definitive policy actions.

This paper is organized as follows. Section 2 presents the modeling framework, including an overview of the model structure and mathematical formulation. Section 3 introduces the scenario analysis implemented to demonstrate the benefits of the approach, including an overview of the assumptions used in the African case study. Section 4 describes results of the case study scenario analysis, and section 5 discusses the main findings and possible future developments. Finally, section 6 summarizes the main conclusions.





Figure 1. Schematic overview of Extended Continental-scale Hydroeconomic Optimization modeling framework. Dashed-lined boxes denote intermediate models or processes used to generate input data, double-lined box represents the optimization module, and the solid box indicates the results. Dashed arrows denote intermediate input data, and solid arrows indicate main input data needed for the optimization. RCP = Representative Concentration Pathway. SSP= Shared Socio-economic Pathway.

2. Modeling Framework

2.1. Model Structure

Figure 1 depicts the main features of the ECHO model including input data and outputs. ECHO is a bottom-up linear optimization model, which includes an economic objective function and a representation of the most relevant biophysical and technological constraints. The objective function of ECHO minimizes the total investment and operating costs of a wide variety of water management options over a long-term planning horizon (e.g., a decade or more), to satisfy sectoral water demands across subbasins within river basins at a continental scale. ECHO optimization approach can be classified as a normative optimization because it goes beyond improvements in the management of existing facilities, toward projections of the capacity and activity levels of various water management options, based on the assumption that water users seek to minimize the cost of water supply and demand management subject to constraints. The optimization procedure in ECHO uses a perfect foresight formulation, which provides the most optimal transition for the water system across the studied spatial and temporal ranges under anticipated future climate, socioeconomic, and policy changes. Most sectoral planning models in the literature assume perfect foresight, although limited foresight in different forms has been applied to some of these models (Keppo & Strubegger, 2010).

The subbasin units are created by intersecting river basin and country administrative boundaries (hereafter *basin-country units* or BCUs) and are linked within a reduced-form transboundary river network. This spatial delineation seeks to cover both the political boundaries of management policies and hydrological domains. Figure 2 shows the spatial delineation used in ECHO for the African case study, which covers 150 BCUs across Africa. However, ECHO is designed to operate at different spatial scales and in different regions or continents, given the data availability. Each BCU is treated as a single unit, meaning that water flows between spatial locations within a BCU are not considered (i.e., water availability is aggregated over a BCU). However,





Figure 2. Delineation of Africa into spatial units and networks in the Extended Continental-scale Hydroeconomic Optimization model: (a) Basins, (b) subbasins, (c) countries, (d) basin-country units (BCUs), (e) subbasin-country units (sub-BCUs), and (f) sub-BCU network. (g) Example identification of the sub-BCU network for the Nile River Basin: the flow accumulation and sub-BCUs are combined to generate a network between sub-BCUs. Network line widths are plotted proportionally to the identified flow accumulation at the outlet and for the example shown connect centroids of the sub-BCUs as opposed to the outlets. Madagascar, Comoros, and Sao Tome and Principe are also included in the analysis but are not depicted here.

water can be transferred between BCUs pertaining to the same river basin, and each BCU can have inflow from upstream BCUs as well as discharge into downstream BCUs and/or a natural sink.

ECHO also includes reasonable representations of essential biophysical and technological features at the BCU level. These include representations of various water supply sources (surface water, groundwater, and nonconventional water such as desalinated water), sectoral demands (irrigation, domestic, manufacturing, and electricity), and infrastructure (surface water reservoirs, desalination plants, wastewater treatment plants, irrigation systems, and hydropower plants). Moreover, ECHO incorporates a continental-scale electricity market for electricity trade between BCUs, which enables the explicit representation of feedbacks between local water constraints and regional electricity development. The GAMS optimization software is used for ECHO development and scenario simulations (Brooke et al., 1988). The optimal solution generated by ECHO provides spatially explicit information on a least cost combination of water management options that can satisfy sectoral water demands looking at water, energy, and land sectors.

2.2. Water Management Options

A diverse range of water management options are represented in ECHO, including supply and demand options that span over the water, energy, and agricultural systems (Table 1). The supply-side management options are surface water diversion, groundwater pumping, desalination, wastewater recycling, and surface water reservoirs. Surface water diversion, groundwater pumping, and desalination all transform raw water resources (surface water, groundwater, and seawater) into freshwater suitable for consumption within the different sectors (irrigation, domestic, manufacturing, and electricity uses). Wastewater recycling enables upgrading of wastewater originating from domestic and manufacturing sources to suitable quality for different purposes. Surface water reservoirs store water across several months for later multipurpose uses. The demand-side management options include different irrigation systems (flood, sprinkler, and drip) and various options to improve crop water management in irrigation and water use efficiency in the domestic and



Table 1

Cost and Performance of Water Management Options

	Electricity	Water	Investment cost ^b		O&M cost ^c			
Options	intensity (KWh/m ³)	efficiency (%) ^a	Unit	Value ^d	Unit	Value ^d	Lifetime (year)	References
<i>Supply-side options</i> Surface water diversion ^e	0.03	90	\$/m ³ /day	57 (34–135)	\$/m ³	0.01 (0–0.05)	10	Albiac et al. (2003), Plappally and Lienhard (2012),
Groundwater pumping ^e	0.1	80	\$/m ³ /day	8.5 (7–15)	\$/m ³	0.01 (0–0.05)	10	Find Stillwein et al. (2010) Fan et al. (2013), Fischer et al. (2007), and Kirchan et al. (2005)
Desalination	3.5	50	\$/m ³ /dav	1700 (900–2500)	\$/m ³	0.25 (0.15-0.35)	30	Ghaffour et al. (2003)
Recycling	1	70	\$/m ³ /day	1300 (600–1700)	\$/m ³	0.15 (0.05–0.25)	30	Iglesias et al. (2010) and Bodriguez-Garcia et al. (2011)
Surface water reservoirs	0		1000\$/Mm ³	450 (150–2500)	\$/m ³	0.05 (0.01–0.3)	60	Keller et al. (2000) and Wiberg and Strzepek (2005)
Flood irrigation ^f	0	60	\$/ha	460 (80–1200)	\$/ha	23 (4–60)	30	Ignaciuk and Mason-D'Croz (2014).
Sprinkler irrigation ^f	0.24	75	\$/ha	650 (200–2800)	\$/ha	33 (10–140)	20	Kahil, Connor, and Albiac (2015),
Drip irrigation ^f	0.18	90	\$/ha	1400 (700–4000)	\$/ha	70 (35–200)	20	Phocaides (2000), and Sauer et al. (2010)
Improved irrigation management ⁹	0	+10% relative to regular management	\$/ha	0	\$/ha	200 (-600-1400)	0	Evans and Sadler (2008), Fischer et al. (2007), Sanchez et al. (2016), and Wada, Gleeson, & Esnault (2014)
Improved domestic and manufacturing management	0	+10% relative to regular management	\$/m ³	0.4 (0.05–1.5)	\$/m ³	0.1 (0.01–0.4)	10	Escriva-Bou et al. (2015) and Rosenberg et al. (2008)

^aWater efficiency measures the relationship between input water (raw water such as seawater or wastewater) and output water (feed water such desalinated water or treated wastewater) for each technology. Estimates of water efficiency by option are taken from Dubreuil et al. (2012). ^bThe investment cost for the supply-side options is expressed in terms of the supply capacity of each option (not in terms of the activity). The investment cost of surface water diversion is estimated based on the cost of different interbasin water transfer projects. ^cO&M cost does not include the energy cost. ^dThe values of both the investment and O&M costs shown are the best estimates with the possible ranges shown in parentheses. ^eThe O&M cost including the energy cost) is assumed to be the same for both surface water diversion and groundwater pumping. ^fFor all irrigation systems, the O&M cost is assumed to be equivalent to 5% of the investment cost. ^gA negative O&M cost indicates that some irrigation management options might be able to concurrently save water and generate additional economic benefits to farmers (i.e., win-win options).

manufacturing sectors. The electricity consumption of these water management options is accounted for, explicitly linking water and electricity systems.

Moreover, several electricity generation technologies are included in ECHO, which convert different forms of primary energy resources into electricity. These technologies consist of a combination of different fuel and turbine types. Thermal power plants are distinguished in particular by types of cooling technology, to enable feedbacks to the water supply system, including once-through and closed-loop cooling systems utilizing freshwater, as well as air-cooled and seawater-cooled once-through systems. Hydropower plants are also included in ECHO and their production depend on the BCU level hydropower generation potential (Fricko et al., 2016; Parkinson, Djilali, et al., 2016).

A significant amount of cost and performance data associated with the water management options considered are required to parametrize ECHO. This information is often available in the literature for only a few regions and can be highly variable depending on factors that include water quality, technology capacity, and political and environmental conditions. Here best estimates of these parameters have been collected by reviewing much of the available literature. These estimates have been scaled for BCUs using information on biophysical (e.g., groundwater table depth) and economic (e.g., energy and labor costs) conditions. The cost and performance parameters of the water management options are provided in Table 1. The full list of the electricity supply technologies implemented in ECHO is provided in Table S1 in the supporting information (SI). ECHO also has the capacity for additional water management options, which can be added when the corresponding cost and performance parameters become available.



2.3. Mathematical Formulation

An overview of the main equations in ECHO is presented in this subsection. In all equations, parameters are represented by lowercase letters and variables are represented by capital letters. A reduced-form water massbalance equation is used in ECHO to ensure water conservation in each BCU $i \in I$ and time step $t \in T$. This equation enables the hydrological connectivity between BCUs in the transboundary river network and can be represented as follows:

$$S_{i,t-}S_{i,t-1} = R_{i,t} + I_{i,t} - Q_{i,t} + RF_{i,t} - D_{i,t} - L_{i,t}$$
(1)

where $R_{i,t}$ is the local runoff and $I_{i,t}$ is the inflow from upstream BCUs in each BCU *i* during time period *t*. $Q_{i,t}$ is the water release to demand sectors, $RF_{i,t}$ is the return flow to the river system, $D_{i,t}$ is the discharge to downstream BCUs and sinks, and $L_{i,t}$ are the evaporation and seepage losses, in each BCU during the same time period. $S_{i,t}$ and $S_{i,t-1}$ represent the storage level of surface water reservoirs at the end of period *t* and t - 1, respectively, in each BCU.

The total inflow from upstream BCUs, *l_i*, *v* in each BCU and time step is calculated as follows:

$$I_{i,t} = \sum_{j} b_{j,i} \cdot D_{j,t} \tag{2}$$

where $D_{j, t}$ is the discharge from each upstream BCU $j \in J$ (with J as a subset of I) and $b_{j, i}$ is a vector of coefficients that links each BCU i to all upstream BCUs j. The coefficients that comprise this vector take on values of +1 if a BCU is linked to an upstream BCU, and 0 otherwise. This linkage between BCUs is defined based on historical discharge data.

The downstream discharge, $D_{i, t}$, in each BCU and time step must be greater than or equal to the minimum environmental flow requirements needed for healthy aquatic ecosystems, $e_{i, t}$, as follows:

$$D_{i,t} \ge e_{i,t}$$
 (3)

For each BCU, ECHO incorporates a supply-demand balance equation, which is tracked for each demand sector $d \in D$ in each time step, so that water withdrawal of each sector must be less than or equal to the total sum of water originating from suitable supply options $s \in S$ (i.e., surface water, groundwater, desalination, and recycling):

$$W_{i,t}^d \leq \sum_{s, s, s, d} Q_{i,t}^s \tag{4}$$

where $W_{i,t}^d$ is the water withdrawal of each sector and $Q_{i,t}^s$ is the outflow from each supply option. The $\gamma_{s, d}$ is a vector that links supply options to demand sectors. The coefficients that comprise this vector take on values of +1 if a demand sector is using a certain supply option, and 0 otherwise. Here we assume that irrigation, domestic, and manufacturing sectors can use all available water supply options, while the electricity sector can only use diverted surface water.

The amount of water withdrawn by each sector is calculated based on the exogenous consumptive water demand of that sector combined with the efficiency of its corresponding demand-side management options (irrigation systems, irrigation management practices, domestic and manufacturing management practices, and cooling technologies; Hanasaki et al., 2018) as follows:

$$\boldsymbol{c}_{i,t}^{d} = \boldsymbol{E}_{i,t}^{d} \cdot \boldsymbol{W}_{i,t}^{d} \tag{5}$$

where the parameter $C_{i,t}^d$ is the exogenous consumptive water demand of each sector and the variable $E_{i,t}^d$ describes the efficiency of the demand-side management options suitable for each sector. For example, the agricultural sector can invest in flood, sprinkler, or drip irrigation systems or different irrigation management practices (as shown in Table 1) with varying levels of efficiency, which affect the quantity of water withdrawn for irrigation and return flows. The volume of return flows from each demand sector, $RF_{i,t'}^d$ is computed as follows:

$$RF_{i,t}^{d} = \left(1 - E_{i,t}^{d}\right) \cdot W_{i,t}^{d} \tag{6}$$

The total amount of return flows to the river system, $RF_{i,t}$, which can be reused by downstream BCUs, is assumed to equal the sum of the return flows from all sectors, excluding domestic and manufacturing return flows that can be recycled and reused within the BCU where they are produced.

A capacity constraint is used to limit the activity of both the supply- and demand-side management option $o \in O$ according to the available physical capacity of the options:

$$Q_{i,t}^{o} \leq Z_{i,t}^{o} \tag{7}$$

where $Q_{i,t}^o$ is the activity level of each management option in each time step, which represents the volume of water supplied by that option. $Z_{i,t}^o$ is the installed capacity of each option in the same time step. The capacity constraint therefore works, for instance, to ensure the volume of desalinated water produced does not exceed the installed desalination capacity or so that the volume of irrigation water supplied via an irrigation system does not exceed the installed capacity of that system.

Moreover, ECHO incorporates capacity expansion decisions $Z_{i,t}^{o,new}$ that alleviate capacity constraints for the different management options. Capacity retirements $Z_{i,t}^{o,ret}$ are further decision variables that allow options to have finite lifecycles. The installed capacity of a particular option is thus given by:

$$Z_{i,t+1}^{o} = Z_{i,t}^{o} + Z_{i,t}^{o,\text{new}} - Z_{i,t}^{o,\text{ret}}$$
(8)

ECHO calculates the total annual cost of supplying water $C_{i,t}^{o,tot}$ by each water management option o. This is equal to the sum of the annual cost of investment in new capacity, the activity operating cost, and the maintenance cost associated with the installed capacity, for each option in each BCU and time step:

$$C_{i,t}^{o,\text{tot}} = \sum_{o} c_{i,t}^{o,\text{inv}} \cdot Z_{i,t}^{o,\text{new}} + \sum_{o} c_{i,t}^{o,\text{OP}} \cdot Q_{i,t}^{o} + \sum_{o} c_{i,t}^{o,\text{Mc}} \cdot Z_{i,t}^{o}$$
(9)

where parameters $c_{i,t}^{o,\text{inv}}$, $c_{i,t}^{o,\text{Op}}$, and $c_{i,t}^{o,\text{Mc}}$ are, per unit, investment cost, activity operating costs, and installed capacity maintenance costs, for each option, respectively.

To determine the optimal solution and the associated decision variables, ECHO minimizes the net present value of the total cost of supplying water in all BCUs at a continental scale over the planning horizon subject to the constraints (1) to (9). Other additional constraints are used, and these are outlined in the following sections describing ECHO's application to Africa. The length of the planning horizon depends upon both the specific problem under consideration and the target objective. The objective function of ECHO takes the following form:

$$\operatorname{Min} C^{\operatorname{npv}} = \sum_{o,i,t} \frac{C_{i,t}^{o,\operatorname{tot}}}{\left(1+\delta\right)^{t}}$$
(10)

where C^{npv} is the net present value of total cost and δ is the discount rate.

3. ECHO Model Application to Africa

The description of ECHO has focused so far on the core modeling framework applicable to different regions or continents. Since we apply ECHO over Africa as a case study, this version of ECHO is hereafter referred to as ECHO-Africa. Africa is a challenging but important region because of its rapidly growing demands and lack of water infrastructure (Cervigni et al., 2015). To highlight the capacity of ECHO model, we examine different scenarios representing alternative socioeconomic and climate futures for Africa toward the year 2050. ECHO-Africa has been solved monthly over the 2010–2050 period in 10-year increments (i.e., five time steps: 2010, 2020, 2030, 2040, and 2050). In this section, we provide an overview of the input data and assumptions used in ECHO-Africa, and we introduce the scenario analysis.

3.1. Spatial Delineation and River Network

Balancing spatial details with computational requirements is critical in ECHO because the size of the optimization problem, as described in the previous section, can increase exponentially with the number of spatial units. Thus, to minimize the computational burden, ECHO-Africa uses the hierarchical spatial delineation depicted in Figure 2. ECHO-Africa runs at the level of BCUs representing the intersection between river basin and country administrative boundaries (Figure 2d). The HydroBASINS data set is used to define the river basin delineation in ECHO-Africa (Figure 2a; Lehner & Grill, 2013). HydroBASINS incorporates subbasins of higher order (Figure 2b), which have a single outlet, but as opposed to river basins, can have inflow from upstream subbasins. ECHO-Africa taps into this feature to track hydrological connectivity between BCUs. Subbasins corresponding to each river basin are initially intersected with national boundary polygons from the Global Administrative Areas Database (2012; Figure 2c) to generate 1622 *subbasin country units* (sub-BCUs; Figure 2e). These sub-BCUs are then dissolved in a subsequent step according to their corresponding basin classification obtaining a consistent set of BCUs (150 BCUs; Figure 2d). Table S2 in the SI provides the list of river basins and countries included in ECHO-Africa.

In the next step, each sub-BCU is overlaid with the gridded flow accumulation data from HydroBASINS at 15 arc sec to identify the main outlet, which is the grid cell within the sub-BCU with the maximum flow accumulation (Figure 2g). Neighboring grid cells located in other sub-BCUs are then checked to see if there exists greater flow accumulation, as this may indicate an overland flow of runoff downstream to that neighboring sub-BCU. Where this is possible, the connection and direction are noted so that the outflow from the upstream sub-BCU can be tracked and accounted for in the analysis of water availability in the downstream sub-BCU. If no downstream grid cell can be identified, the sub-BCUs, resulting in a reduced-form transboundary river network that distinguishes subbasins by country. An example of the network generated for the Nile River Basin is depicted in Figure 2g. For the spatial delineation of Africa, the continental network has 1,622 nodes and 1,031 links (Figure 2f).

The total average monthly runoff in each sub-BCU is estimated using the rainfall-runoff component from the global hydrological model PCR-GLOBWB (van Beek et al., 2011; Wada, Wisser, & Bierkens, 2014). These monthly runoff estimates act as nodal inputs to the network. The sub-BCU network database is linked with a network analysis tool to enable efficient identification of upstream nodes from any point in the network (Csardi & Nepusz, 2011). This facilitates tracking of natural flows in the network, where any unmanaged nodal inputs are assumed to accumulate in downstream nodes. The network between sub-BCUs is further simplified for the optimization procedure in ECHO to represent a network between BCUs. In order to account for environmental flow requirements in the model simulations (as shown in equation (3)), we assume a minimum outflow of 30% of monthly runoff in each BCU, following the study by Cai and Rosegrant (2002).

3.2. Existing Capacity of Water Management Options

In order to estimate future investment needs, first the existing capacity of the different water management options implemented in ECHO is assessed at the BCU level. Here we gather information on existing capacities from various databases. The capacities of existing surface water reservoirs are estimated by aggregating facility-level data from the GRanD database (Lehner et al., 2011). Evaporative losses due to increased surface area during reservoir storage are incorporated into the water mass-balance equation defined in section 2.3 using a linearized area-volume relationship (Lele, 1987). New future investments in surface water reservoir capacity are constrained within each BCU where reservoir exclusion zones were distinguished (Liu et al., 2018). The existing capacities of surface water diversion and groundwater pumping infrastructure are identified using historical gridded water withdrawals and groundwater extraction rates from Wada et al. (2010, 2011). These withdrawals are aggregated to the level of the BCUs, and the maximum monthly withdrawal in the historical time series plus a 10% reserve margin is used to define the capacity in each BCU. Moreover, the global hydrological model PCR-GLOBWB (van Beek et al., 2011; Wada, Wisser, & Bierkens, 2014) includes groundwater recharge as well as baseflow component in the groundwater balance and connects baseflow discharge from groundwater to the river systems. In this paper, estimates of monthly groundwater recharge at grid scale are used to compute the availability of renewable groundwater resources in each BCU. Therefore, ECHO calculates a groundwater budget in each time step and BCU and provides the level of depletion of groundwater resources, as the difference between pumping and renewable groundwater resources.

Existing desalination capacities are identified using a refined version of the global desalination database (Global Water Intelligence, 2017). Wastewater treatment capacities are defined using estimates of return

flows from the domestic and manufacturing sectors and national data on water treatment access rates from Baum et al. (2013). For countries without historical data, the water treatment access rate is estimated by matching each country to another with similar gross domestic product (GDP) per capita. The existing water treatment capacity is estimated in each BCU by multiplying the estimated water treatment access rate for 2010 by the maximum volume of domestic and manufacturing return flows. Figure S2 in the SI depicts the estimated capacities of selected water management options in 2010.

3.3. Sectoral Water Demands

Monthly sectoral water demands for the period 2010-2050 at BCU level are estimated to be included as inputs into ECHO-Africa. This section provides an overview of the estimation procedures and data sources. Monthly domestic water demands are estimated following the approach used in several previous studies (Hanasaki et al., 2013; Hejazi et al., 2014; Wada et al., 2016), which use harmonized projections of population, GDP, and technological change from the Shared Socioeconomic Pathways (SSPs) database (O'Neill et al., 2014). Manufacturing water demands are estimated following an approach similar to one reported in Hejazi et al. (2014). To estimate the historical water demand of the manufacturing sector at the country level, data on the historical water demands of the industrial sector (manufacturing plus electricity) are taken from AQUASTAT database (Food and Agriculture Organization, 2016) and reduced by the 2010 water demand of the electricity sector, as calculated in ECHO-Africa (following the approach described in section 3.4). Future changes in manufacturing demands are then projected using a statistical log-linear model fit to the historical GDP and manufacturing data (Parkinson, Johnson, et al., 2016). National manufacturing water demand projections are then downscaled to BCU level using the gridded urban population projections aligned with the SSPs from Jones and O'Neill (2016). The volume of return flows from both the domestic and manufacturing sectors is determined by recycling ratios developed per country taken from Wada et al. (2011) and Wada, Wisser and Bierkens (2014).

Monthly irrigation water demands for the period 2010–2050 are estimated at BCU level using irrigated crop area and monthly gross water requirements per unit area. In order to estimate irrigated crop area in each BCU for the period 2010–2050, data on historical (year 2000) cropping patterns for major irrigated crops at the global scale with a spatial resolution of 5 min are obtained from the MIRCA2000 data set (Portmann et al., 2010). This gridded crop area is aggregated across each BCU, and crop distributions (i.e., the fraction of each crop area relative to total irrigated area) are calculated. These historical crop distributions are combined with country-specific projections of future change in irrigated area from 2000 to 2050 according to SSP scenarios originating from the Global Agro-eccological Zones model (Fischer et al., 2012). This provides a first-order estimate of future irrigated area at BCU level. This method is unable to reproduce changes in crop distribution within BCUs but adequately reflects the large-scale dynamics of the expanding irrigated areas over the coming decades (Wisser et al., 2010).

Net water requirements for irrigation per unit crop area (i.e., consumptive demands) are estimated using the crop coefficient method (Allen et al., 1998). Monthly crop evapotranspiration is calculated by combining a crop coefficient per crop development stage with a monthly reference (potential) evapotranspiration for the period 2010–2050. Net monthly irrigation requirements are calculated at BCU level, so as to ensure the optimum growth of each crop. These net requirements are the difference between crop evapotranspiration and actual evapotranspiration. Crop-specific calendars and growing season lengths are obtained from the MIRCA2000 data set, while crop coefficients and potential and actual evapotranspiration are taken from the PCR-GLOBWB model (van Beek et al., 2011; Wada, Wisser, & Bierkens, 2014). Lastly, irrigation requirements and irrigation efficiency. This efficiency factor measures the overall effectiveness of irrigation, which takes into account losses during water conveyance as well as application efficiency of irrigation systems (flood, sprinkler, and drip) at plot level, and management practices (regular or improved).

3.4. Agriculture and Energy System Linkages

To integrate the agricultural and water systems, ECHO includes investments in irrigation systems and management practices as decision variables. These investment decisions have impacts on the volume of irrigation water withdrawn from the water system and the cost of water supply to the agricultural system. The current irrigation efficiency level and irrigation system distribution in each BCU for the model base year (2010) are assumed to be similar to the corresponding country-level data as reported by the AQUASTAT database (Food and Agriculture Organization, 2016). The maximum theoretical efficiency of irrigation systems at plot level is taken from Phocaides (2000) and further adjusted in order to match with the current overall irrigation efficiency in each BCU. Because not all crop types can be irrigated using all irrigation systems (for instance, paddy rice needs flood irrigation and some crops cannot use sprinklers), ECHO-Africa includes a constraint that defines the suitability between irrigation systems and crop types following Sauer et al. (2010). Moreover, the inclusion of irrigated crop area for each BCU enables additional water management options and interactions between the agricultural and water systems to be integrated into ECHO-Africa in future model developments. These might include the possibility of virtual water trade.

To account for feedbacks between local water constraints and regional electricity development, ECHO-Africa includes a reduced-form electricity sector planning module following an approach similar to one reported in Loulou and Labriet (2008) and Loulou (2008). The objective of the module is to satisfy the monthly electricity demand (including non-water- and water-related demands) in each BCU during the model's planning horizon at a minimum supply cost (including investment and operating costs) subject to various technical and resource constraints, by simultaneously making decisions on investment and operation of various electricity supply technologies (as shown in Table S1 in the SI). The main constraints included in the electricity module are the availability of primary energy resources, the availability of physical capacity of technologies (which accounts for existing, new, and retired capacities in a similar way as equation (8) in section 2.3), supply-demand balance, and peak load constraint (which is used to insure against unexpected events such as unplanned technology down time or random peak demand that exceeds the average demand).

BCUs that pertain to the same power pools can trade electricity, as delineated and parameterized in Taliotis et al. (2016). This links electricity generation in the BCUs at the continental scale. Electricity transmission is modeled using a simple transport representation between power pools and does not address voltage and power flow constraints. Table S3 in the SI depicts the list of countries for each of the five major African power pools. The electricity demand of nonwater activities are taken from the study by Bauer et al. (2017), which projects the future of the energy sector using an ensemble of integrated assessment models and harmonized projections of population, GDP, and technological change from the SSPs database. The electricity demand of water-related activities is calculated endogenously in ECHO-Africa using the electricity systems. The electricity module is calibrated using data on electricity production and technology mix at country level from United Nations (2017).

Moreover, ECHO-Africa includes additional relevant connections between water and electricity systems. Hydropower is one such connection and is assessed in each BCU following the approach described by Cuya et al. (2013). The approach tracks water inflows in conjunction with elevation changes between the inlets and outlets at the BCU level to estimate total hydropower potential. Hydropower generation is then calculated using a linear model that is fitted between historical total inflow and hydropower production in each BCU and constrained by estimated hydropower potential following the approach described by Khan et al. (2017). Thermal power plant cooling technologies also represent an important electricity-water linkage and are modeled explicitly in ECHO-Africa. Each thermal power plant type is defined in terms of its cooling technology type (once-through and closed-loop cooling systems utilizing freshwater, air-cooled system, and seawater-cooled once-through systems). Existing cooling technology capacities for 2010 are estimated using cooling technology shares compiled for each BCU from a facility-level data set described by Raptis and Pfister (2016). The water demand of the electricity sector is thus calculated endogenously in ECHO-Africa using the volume of water withdrawn by each cooling technology and its optimized activity level following the approach described by Fricko et al. (2016).

3.5. Socioeconomic and Climatic Scenarios

A set of global water scenarios based on combinations of the Shared Socioeconomic Pathways (SSPs) and Representative Concentration Pathways (RCPs) have been developed by Wada et al. (2016), in order to represent the full range of plausible and anticipated future climatic and socioeconomic changes. These scenarios have been used to assess future water scarcity conditions using hydrological models at both the global (Wada et al., 2016) and the regional (Satoh et al., 2017) scales. In this paper, we use these scenarios in order to explore strategies aimed at balancing future water demand and availability in Africa using ECHO-Africa for



Table 2

Summary of the Socioeconomic and Climatic Scenarios

		_	Water demand			
Scenario	Water availability	Domestic	Industrial	Irrigation	Constraints	References
MoR	Runoff for the period 2010–2050 is projected using the global hydrological model PCR-GLOBWB, with the climate forcing data based on RCP 6.0 scenario.	Water demand of th period 2010–2050 b population growth a for the SSP2 scenario	e different sectors are ased on assumptions and technological dev o.	e projected for the about GDP growth, elopment,	Groundwater pumping can increase by up to 50% compared to historical maximum pumping in all BCUs for the period 2020–2050. A minimum outflow constraint of 30% of runoff in each BCU for the period 2010–2050 is implemented. A minimum use constraint for existing expensive water options such as desalination, recycling or efficient irrigation system is implemented. A constraint prioritizing the use of desalinated water over other sources (that are less expensive) for the domestic and manufacturing sectors in coastal bacins is implemented.	Wada et al. (2016), Nasta et al. (2016), Cai and Rosegrant (2002), Hanasaki et al. (2018), and United Nations (2017)
RR	Runoff decreases by 10% in each decade compared to MoR for all BCUs in the period 2020–2050.	Increases by 100% in 2050, +25% per decade, compared to MoR, for all BCUs in the period 2020–2050.	Increase by 80% in 2050, +20% per decade, compared to MoR, for all BCUs in the period 2020–2050.	Increases by 20% in 2050, +5% per decade, compared to MoR, for all BCUs in the period 2020–2050.	A constraint defining the share of hydropower in the electricity technology mix for the period 2010–2050 is implemented.	
Sust	Runoff increases by 10% in each decade compared to MoR for all BCUs in the period 2020–2050.	Decreases by 50% in 2050, -12.5% per decade, compared to MoR, for all BCUs in the period 2020-2050.	Decrease by 40% in 2050, -10% per decade, compared to MoR, for all BCUs in the period 2020-2050.	Decrease by 20% in 2050, -5% per decade, compared to MoR, for all BCUs in the period 2020-2050.		

Note. BCU = basin-country unit; GDP = gross domestic product; SSP = Shared Socioeconomic Pathway; RCP = Representative Concentration Pathway; MoR = Middle of the Road; RR = Regional Rivalry; Sust = Sustainability.

the period 2010–2050. Three alternative scenarios are used here, which encompass a higher, middle, and lower range of plausible future changes in climate and society toward 2050 (Table 2). Our objective is thus to demonstrate ECHO's potential across contrasting scenarios, rather than to recommend definitive policy actions (i.e., a sensitivity analysis).

The middle-range scenario (hereafter referred to as *Middle of the Road* [*MoR*]) represents a combination of SSP2 and RCP6.0, assuming relatively moderate changes in socioeconomic and climatic drivers. All assumptions (e.g., GDP and population growth rates, and technological changes) used in this scenario vary among BCUs. For the other two scenarios, the assumptions concerning relative changes to water demand and availability compared to MoR scenario are applied uniformly for all BCUs, but they are still consistent with possible overall changes in Africa. The higher-range scenario (hereafter referred to as *Regional Rivalry* [*RR*]) represents a scenario with slow economic growth, large population growth, and negative climate change impacts on water availability (corresponding to SSP3). The lower-range scenario (hereafter referred to as *Sustainability* [*Sust*]) represents a scenario with medium-high economic growth, low population growth, and positive climate change impacts on water availability (corresponding to SSP1).

To assure that ECHO-Africa is producing robust future projections, estimated water withdrawals of all sectors at country level for the base year 2010 have been validated against the data available in AQUASTAT and calibrated, if needed, to correct data imperfections and get the baseline solution close to the observed values.

The calibration procedure involved adjustments in some model parameters such as gross crop water requirements or irrigation efficiency. Per capita water use for drinking water has been already taken from the reported country statistics from AQUASTAT. Moreover, some variables in ECHO including the installed capacity of all water management options and the share of the different supply sources (surface water, groundwater, desalination, and recycling) relative to total withdrawals are all fixed at observed levels for the base year 2010. For the rest of the planning horizon (from 2020 to 2050), a constraint that assures a minimum use of existing capacity of the water management options is implemented in order to force the optimal solution to be within the physical and logical system boundaries. Additional constraints are also used such as prioritizing the use of desalinated water over recycled wastewater or groundwater for the domestic and manufacturing sectors in coastal BCUs and the definition of the share of hydropower in the electricity technology mix in each BCU. This approach is widely used in large-scale modeling, given the lack of detailed local-level information for calibration (Hanasaki et al., 2018). The results of the validation procedure are provided in Table S4 in the SI.

4. Results

In this section, we present the results of ECHO-Africa simulations for the three alternative scenarios. To keep the scope of the result subsections within reasonable limits, we focus on results aggregated to the continental-scale and some relevant regional findings. We also focus on the end of the simulation period (the year 2050) and compare it to the base year (2010).

4.1. Least Cost Combination of Water Management Options

Figure 3 depicts total water demand, sectoral water withdrawals, and the sources of water, aggregated to the continental scale for 2010 and 2050, and the 10 countries and basins with the highest changes in withdrawals over the simulation period, across the three alternative scenarios. Figures 4 and 5 provide results of water withdrawals at the BCU level by sector and source, respectively. Figures S3 to S6 in the SI provide the same results aggregated to basin and country scales. Results from Figure 3 show that total water withdrawals in Africa amount to 460 km³/year in 2010 and are projected to reach between 560 and 820 km³/year by 2050 depending on the scenario, an increase of 20–80% compared to current withdrawals. As expected, the RR scenario has the largest increase in withdrawals driven by the highest population growth. This is followed by the MoR scenario and the Sust scenario, respectively. The largest increases in water withdrawals in 2010 and 2050 in all scenarios are expected to take place in areas with already large withdrawals in 2010 such as in Egypt by country, followed by Ethiopia and Nigeria, and in the Nile by basin, followed by Niger and Congo. However, some areas with marginal water withdrawals in 2010 may increase significantly their withdrawals by 2050 such as the Democratic Republic of the Congo, Tanzania and Burundi by country, and Africa West Coast, Orange, and Lake Chad by basin.

The increase in total water withdrawals is primarily driven by growing manufacturing and domestic withdrawals, which are projected to rise substantially from 38 km³/year in 2010 to 90–279 km³/year in 2050. Moreover, the growth in electricity generation in Africa will be considerable, with water withdrawals for cooling of thermal power plants projected to increase significantly, from 1 km³/year in 2010 to 18–30 km³/year in 2050. Most increases in nonirrigation withdrawals are found in the Congo, Lake Chad, and Niger by basin and in the Democratic Republic of the Congo, Tanzania, and Burundi by country (Figures 4, S3, and S5 in the SI). At the same time, irrigation withdrawals are expected to continue increasing between 2010 and 2050, but at a slower pace, induced by both climate change impacts and growing food demand, from 420 km³/year in 2010 to 473–576 km³/year in 2050. Irrigation will remain the largest water user in Africa, but its relative share is projected to decrease by 2050. Major increases in irrigation withdrawals are mostly found in the Niger, Nile, and Volta by basin and in Uganda, Nigeria, and Mozambique by country (Figures 4, S3, and S5 in the SI). It is important to note that the assumptions related to the changes in irrigated areas and cropping patterns used in ECHO may overstate the gains from improved irrigation efficiency, and therefore, estimated irrigation withdrawals should be considered as part of scenario assessments, which could be further improved subject to improved spatial data sets in irrigation area and distribution.

Results from Figure 3 show that various water sources are used to satisfy the water withdrawals. Surface water is the major source of water in all scenarios. Surface water diversion increases from 390 km³/year in 2010 to





Figure 3. (a) Total water demand, actual sectoral water withdrawals, and water sources in the African continent for each scenario. (b) Ten countries (left column) and basins (right column) with highest change in withdrawals between 2010 and 2050 in Africa for each scenario. Total water demand is estimated for 2050 by assuming no change in baseline efficiency across all sectors. This highlights water conservation originating from the implementation of demand-side management options. Nonconventional water includes both desalinated water and recycled wastewater. Full names for countries and basins are provided in Table S2 in the SI.

460–590 km³/year in 2050 (an increase of 18–50%), driven by an increase in surface runoff in some basins or a better allocation of water over time and space. The largest increases in surface water withdrawals across all scenarios are found in the Congo basin, followed by the Lake Chad and the upstream part of the Nile basin, and in several countries in sub-Saharan and southern Africa (Figures 5, S4, and S6 in the SI). Groundwater pumping also increases from 70 km³/year in 2010 to 100–150 km³/year in 2050 (an increase of 43–114%). These increases are projected to take place in all scenarios mostly in northern Africa including the Mediterranean South Coast, Africa North West Coast, and Africa North Interior by basin and Egypt, Algeria, and Tunisia by country, and in southern Africa including the Orange, Africa South Interior, and South Africa South Coast by basin and South Africa, Namibia, and Botswana by country (Figures 5, S4, and S6 in the SI). The use of nonconventional water (including both desalination and recycling) in Africa stood at about 1 km³/year in 2010, or less than 0.3% of the 2010 supply mix. For the MoR and Sust scenarios, the quantity and share of nonconventional water remain almost unchanged. However, the use of nonconventional water remain almost unchanged. However, the use of nonconventional water remain almost unchanged. However, the use of nonconventional water remain almost unchanged. However, the use of nonconventional water remain almost unchanged. However, the use of nonconventional water remain almost unchanged. However, the use of nonconventional water remain almost unchanged. However, the use of nonconventional water remain almost unchanged. However, the use of nonconventional water grows considerably to support accelerated demand growth in the RR scenario, reaching 30 km³/year in 2050, or 4% of the 2050 supply mix. An expansion of desalination capacity under the RR scenario is projected in





Figure 4. Sectoral water withdrawals at the BCU level in 2010 and percentage change of withdrawals in 2050 compared with 2010 for each scenario. The boundaries of the BCUs are highlighted with thin gray lines while the boundaries of the 28 major basins in Africa are highlighted with thick black lines. The list of BCUs and major basins is provided in table S2 in the SI. BCU = basin-country unit; MoR = Middle of the Road; RR = Regional Rivalry.

several coastal areas including Africa West Coast, Gulf of Guinea, Namibia Coast, and Mediterranean South Coast by basin and Egypt, Namibia, and Angola by country. Similarly, an expansion of recycling capacity under the same scenario is projected in many areas such as the Congo, Mediterranean South Coast, and Volta by basin and Mali, South Africa, and Sudan by country (Figures 5, S4, and S6 in the SI).

Results from Figure 3 show also the benefit of using demand-side management options, which allow for the conservation of water resources. These include improving irrigation efficiency by adopting efficient irrigation systems such as sprinkler and drip systems. Domestic and industrial water may also be better managed through the adoption of water saving equipment in industrial plants and households, the reduction of leak-age in water distribution networks, and switching to more water-efficient cooling technologies in thermal





Figure 5. Water withdrawals by source of water at the BCU level in 2010 and percentage change of withdrawals in 2050 compared with 2010 for each scenario. The boundaries of the BCUs are highlighted with thin gray lines, while the boundaries of the 28 major basins in Africa are highlighted with thick black lines. The list of BCUs and major basins is provided in Table S2 in the SI. BCU = basin-country unit; MoR = Middle of the Road; RR = Regional Rivalry.

power plants. Indeed, without implementing the demand-side management options, total water demand in Africa would reach 650–980 km³/year by 2050, an increase of 15–20% compared to actual withdrawals expected in 2050. The rate of adoption of the different demand-side management options varies among BCUs and scenarios, but a higher adoption rate is necessarily found in the RR scenario. A more detailed description of the adoption of the demand-side management options is provided in the SI.

4.2. The Cost of Water Management Options

Figure 6 depicts the annual discounted total cost (including both investment and operating costs) of water supply aggregated to the continental scale (hereafter referred to as the water supply cost) from 2010 to





Figure 6. (a) Annual water sector cost in 2010 and 2050 by scenario and management option for the African continent. (b) Ten countries (left column) and basins (right column) with highest change in annual cost between 2010 and 2050 by scenario. Full names for countries and basins are provided in Table S2 in the SI. MoR = Middle of the Road; RR = Regional Rivalry.

2050, and the 10 countries and basins with the highest changes in the water supply cost over the simulation period, for the three alternative scenarios. This water supply cost comprises the costs of supplying water from different water sources (including surface water diversion, groundwater pumping, nonconventional water production, and reservoirs), the cost of demand-side management options in irrigation, domestic and manufacturing sectors, and the cost of cooling in thermal power plants. Nevertheless, it is important to note that our cost assessment does not account for all the potential costs associated with the water management options implemented in ECHO (e.g., transaction and environmental costs).

Results from Figure 6 indicate that the water supply cost in 2010 amounts to 67 billion USD/year (roughly 3% of the continent's current GDP) and will increase over the coming decades, regardless of the scenario. This cost is projected to increase by 25% in 2050 compared to the present condition in the MoR scenario, reaching 83 billion USD/year. Following a sustainable pathway (Sust scenario) would likely reduce the cost by 15% (or 12 billion USD) in 2050 compared to the MoR scenario. However, following a less sustainable pathway (RR scenario) could result in a significant increase of the cost by 60% (or 50 billion USD) in 2050 compared to the MoR scenario. Interestingly, these results suggest that the cost increase under the RR scenario exceeds the savings made under the Sust scenario. This arises when the low-cost options such as surface water diversion and groundwater pumping reach full capacity as water demand continues to increase, and therefore, ECHO starts expanding existing capacities and investing gradually in more expensive options such as efficiency measures and nonconventional water production.

Figure 6 shows that a major part of the water supply cost originates from the operation and expansion of surface water diversion, groundwater pumping, and surface water reservoirs (66 billion USD or 99% of the



Table 3

Existing Estimates of the Cost of Water Supply and Adaptation to Future Scenarios

Study	Objective of the study	Spatial scale	Methodology	Cost estimate
Briscoe (1999)	Estimating 1990 spending on water infrastructure	All developing countries worldwide	Literature review	65 billion USD/year
Woetzel et al. (2016)	Estimating current and future (year 2030) spending on water infrastructure	Global	Literature review	200 billion USD/year in 2016 500 billion USD/year in 2030
Kirshen (2007)	Estimating adaptation costs for two scenarios of socioeconomic and climatic changes (Intergovernmental Panel on Climate Change scenarios B1 and A1b)	200 countries around the world including many African countries	Simple unit cost estimates	Additional 130–140 billion USD/year by 2030 compared to 2000 for Africa
Ward et al. (2010)	Estimating the cost of adaptation to climate change for the industrial and municipal water supply sectors	Global including Africa	Intervention-based needs assessment	19 billion USD/year for developing countries (3–6 billion USD/year for Africa) for the period up to 2050
Schmidt- Traub (2015)	Estimating the investment cost for ensuring access to safe water and improved sanitation including the incremental costs for dam construction and flood protection	Global	Literature review	49 billion USD/year for the period 2015–2030, with major investments needed in low and lower-middle income countries

total cost in 2010 and 66-80 billion USD or 60-90% of the total cost in 2050, depending on scenario). The cost of demand-side management options (or efficiency measures) increases over time in all scenarios, representing 6% (4-5 billion USD) of the total cost in 2050 under both the MoR and Sust scenarios, and 18% (24 billion USD) of the total cost in 2050 under the RR scenario. The cost of nonconventional water increases moderately over time under the MoR and Sust scenarios to amount to 0.4–0.7 billion USD in 2050 (less than 1% of the total cost), respectively. However, it increases considerably under the RR scenario, reaching 25 billion USD in 2050 (19% of the total cost). The cost of cooling thermal power plants is negligible under both the MoR and Sust scenarios (2-4 million USD by 2050), while it rises sharply under the RR scenario, reaching about 1 billion USD by 2050 (although its share remains less than 1% of the total cost). Results from Figure 6 also indicate that potential future socioeconomic and climatic changes will likely have different implications on the water supply cost in different areas. The highest increases in the water supply cost between 2010 and 2050 for all scenarios are found by country in Nigeria, Egypt, South Africa, Algeria, Morocco, and Democratic Republic of the Congo and by basin in the Nile, Mediterranean South Coast, Africa West Coast, Niger, and South Africa South Coast. The different cost implications are obviously driven by the different changes in total water withdrawals and also by the different choices of water management options. For instance, the change in the cost of water supply between 2010 and 2050 in the Mediterranean South Coast basin is higher than that of the Congo basin, despite the much larger change in total water withdrawals in the latter. This is mainly because more expensive management options, such as supply expansion with nonconventional water sources and use of pressurized irrigation systems, need to be implemented in the Mediterranean South Coast basin to address the growing water demand.

Our cost estimates can be benchmarked against various existing estimates from previous studies of the cost of adaptation to the impacts of future socioeconomic and climatic changes on water resources at the global scale including some estimates for Africa, as shown in Table 3. However, it is important to note that the meth-odological approaches, input data, spatial and temporal resolutions, and scenario assumptions differ between these studies, and therefore, direct comparison of results is not straightforward. Despite the differences between our study and those in Table 3, it appears that our cost estimates are broadly consistent with previous estimates. For example, our 2010 water supply cost in Africa is comparable with the overall spending on water-related infrastructure provided by Briscoe (1999) for developing countries and stands within the same order of magnitude as the global spending estimates provided by Woetzel et al. (2016). Our 2050 water supply cost in Africa aligns with the estimates of Ward et al. (2010) for Africa but is lower than the investment range calculated for Africa by Kirshen (2007), whose study was conducted at country level and using only supply-side management options. Lastly, our cost estimates are consistent with estimates to achieve water-related Sustainable Development Goals (SDG6) globally (Schmidt-Traub, 2015), which could be considered an upper bound cost.



Figure 7. (a) Electricity use of the water sector in the African continent in 2010 and 2050 by scenario and management option. (b) Ten countries (left column) and basins (right column) with highest change in electricity use of the water sector between 2010 and 2050 by scenario. Full names for countries and basins are provided in Table S2 in the SI. MoR = Middle of the Road; RR = Regional Rivalry.

Despite the cost implications, adaptation of the water resources system to future socioeconomic and climatic changes may involve trade-offs among various environmental and economic objectives. For instance, some of the identified adaptation options in this study may be inconsistent with climate change mitigation targets because they involve high levels of energy consumption, such as desalination, recycling, pumping, and pressurized irrigation systems. Indeed, in the RR scenario, we find that electricity use in the water sector could increase fivefold (or by 125 TWh) by 2050 compared to 2010. The largest increase originates from the development of nonconventional resources, which accounts for more than 40% of water-related electricity use in 2050 (only 14% in 2010; Figure 7). Major increases in electricity use between 2010 and 2050 in all scenarios are found by country in Sudan, Egypt, Nigeria, Burkina Faso, Ethiopia, and Madagascar and by basin in the Nile, Volta, Madagascar, Africa West Coast, Gulf of Guinea, and Mediterranean South Coast. However, significant potential for energy savings in the water sector of these basins remains, for example, by tapping the energy embedded in wastewater and reducing water losses along the supply chain (International Energy Agency, 2016).

5. Discussion

In this paper, we present the development of the ECHO model. ECHO was applied to the challenging continent of Africa in order to assess the scope of possibilities to adapt to various scenarios of socioeconomic and climatic changes, while at the same time considering the interactions between water, energy, and agricultural systems. The results of this study highlight the importance of water resources in Africa for future human and economic developments. These developments are projected to heighten water withdrawals across the continent, with potential consequences on available freshwater resources and economic costs, especially in the current water scarcity hot spots such as northern Africa basins, the downstream part of the Nile basin and several basins in southern Africa. Importantly, results show that the varying scenario assumptions lead to different combinations of supply and demand-side management options in different areas, which demonstrate ECHO's sensitivity to changes in parameters and therefore make ECHO eligible for providing more effective policy support.

Furthermore, this study also provides important insights on the spatial differences in water supply and demand patterns among the different African countries and basins and highlights opportunities for more strategic and efficient water management options. Such information could be useful for global donors and policymakers in order to make more informed and impactful investments in water-related sectors across the whole Africa. As an illustration of the benefit of this large-scale HE modeling analysis, our results show that different basins in Africa choose different options for reducing water scarcity, which are appropriate and cost effective for each basin. For example, the growing water demand in the coastal basins of North Africa (e.g., Mediterranean South Coast and Africa North West Coast) throughout the simulation period in all scenarios will likely be covered by expanding the supply of groundwater resources and nonconventional water resources (i.e., desalination and recycling; Figure 5). Increased groundwater withdrawal in these basins is mostly unsustainable (as shown in Figure S7 in the SI), which can have adverse economic and environmental consequences (Raskin et al., 1997). Further expansion of irrigation for food production in those basins appears to be unfeasible from a water resources perspective, and more sustainable management options need to be implemented to reduce groundwater depletion in these basins. These include, for example, improving irrigation efficiency and growing more drought-resistant crops, together with targeted measures, such as metering and regulation of groundwater pumping. Alternative food trade strategies may also be needed, given that most unsustainable groundwater withdrawal in these basins is used to irrigate export crops (Dalin et al., 2017). In contrast, in the Congo basin in sub-Saharan Africa, most demand increase will be covered by the plentiful surface water resources. Irrigation withdrawal in the Congo basin remains relatively small compared to available freshwater resources and could be potentially further expanded in the future. This example of spatial differences in water supply and demand patterns among different African basins indicates the existence of considerable opportunities to improve the sustainability of water use in the whole continent through the implementation of more strategic and efficient management options such as the possibilities for food trade (or virtual water transfer), enhanced electricity trade, or water transfers between basins across the continent.

A certain number of simplifying assumptions were used in defining the structure of ECHO model. ECHO is a linear optimization model incorporating a simplified representation of complex water resources systems. In ECHO, we assume that the objective of water resources systems is to minimize the cost of water supply, given a set of constraints. However, real water resources systems are managed according to broader criteria, including reliability, environmental performance, and energy use. Other important assumptions used in ECHO are associated with the aggregation of spatial units. While these assumptions are reasonable for modeling at large scale, it may not be suitable for detailed basin studies, in which the spatial distribution of water supply and demand may need to be represented explicitly. Moreover, given the lack of reliable biophysical and economic information, this study does not thoroughly consider important local processes including environmental flow requirements (i.e., simple representation in the current version of ECHO), impacts on water quality, impacts of groundwater overexploitation, and the potential increase of water supply costs over time. These processes need detailed local information to capture the impacts at a large scale and the inherent uncertainty in existing large-scale calculations may need further consideration to incorporate in this type of large-scale HE models. For example, groundwater overexploitation can have considerable negative impacts at local scale, such as increased pumping costs, reduced base flow, saltwater intrusion, and land subsidence.

Another limitation of this study is related to our findings on the water saving potentials of improving irrigation efficiency, which differ from those of several previous studies indicating that improved irrigation efficiency does not necessarily save water at basin scale, but it could rather foster increased water use and depletion driven by the expansion and intensification of irrigation activities. This could lead to the reduction of return flows, which could be essential for downstream consumptive and environmental uses subject to water quality issues (Perry & Steduto, 2017; Ward & Pulido-Velazquez, 2008). The reason for these different findings is that our study assumes improvements in irrigation efficiency with current spatial crop area and distribution (Berbel et al., 2018). More dynamic land use optimization module will be included in a next version of ECHO to be able to identify such effects. Lastly, the optimization procedure in ECHO uses a perfect foresight formulation, which might lead to unjustified optimism about the efficiency of water management options (e.g., unrealistic reservoir operation with large (little) carryover storage prior to drought (wet) years or an overvaluation (underestimation) of the value of existing (new) investments), given that in reality decision makers do not have full information about future changes.

Despite all these limitations, ECHO currently serves well for a comparative analysis of the effects of different scenarios and management options, since the errors and limitations are applied uniformly across model runs. Future work will aim to improve the model structure, incorporating additional important biophysical processes such as groundwater dynamics and environmental flow requirements, using more disaggregated spatial representations and local-level data, improving model calibration and validation, and using different optimization procedures. Additional water management options and local practices such as controlling water demand with, for example, the possibility to adjust crop patterns and deficit irrigation, virtual water trade, water transfers, and green infrastructure will be also investigated and included in future works. To provide further insight into robust water management strategies, future work should also consider representations of uncertainties underlying the key model parameters.

6. Conclusions

In this study, we present the development of a new large-scale HE model (ECHO), which fully integrates biophysical, technological, and economic features of water resources systems. ECHO covers multiple subbasin units interacting at continental scale within a reduced-form transboundary river network and involves the main water users at subbasin level. The embedded linkages between subbasin units and sectors at continental scale in ECHO provide a unique opportunity to model water management options at multiple spatial scales and account for their impacts on energy and agricultural sectors. ECHO was applied over Africa with the aim of demonstrating the benefits of this integrated HE modeling framework. Results of this application were found to be consistent with previous studies assessing the cost of water supply and adaptation to future socioeconomic and climatic changes in Africa. Moreover, the results provide insight into several critical areas related to future investments in both supply- and demand-side management options, the varying implications of contrasting future scenarios, and the potential trade-offs among economic and environmental objectives. Overall, results highlight the capacity of ECHO to address challenging research questions related to the sustainable supply of water and the impacts of water management on energy and food sectors and vice versa. As such, we propose ECHO as useful tool for water-related scenario analysis and policy options evaluation.

Acknowledgments

The authors acknowledge the Global Environment Facility (GEF) for funding the development of this research as a part of the Integrated Solutions for Water, Energy, and Land (ISWEL) project (GEF Contract Agreement: 6993) and the support of the United Nations Industrial Development Organization (UNIDO). The authors also acknowledge the continuous support from the Austrian Development Agency (ADA) to the Water Futures and Solutions (WFaS) initiative at Water Program of IIASA. We wish to thank anonymous reviewers whose comments helped substantially improve and clarify this manuscript. The data used are listed in the references, tables, figures and supporting information.

References

- Acquah, S., & Ward, F. A. (2017). Water policy interventions for food security in Afghanistan. *International Journal of Water Resources Development*, 1–22. https://doi.org/10.1080/07900627.2017.1399108
- Albiac, J., Uche, J., Valero, A., Serra, L., Meyer, A., & Tapia, J. (2003). The economic unsustainability of the Spanish National Hydrological Plan. International Journal of Water Resources Development, 19(3), 437–458. https://doi.org/10.1080/0790062032000122961
- Alcubilla, R. G., & Lund, J. R. (2006). Derived willingness-to-pay for household water use with price and probabilistic supply. Journal of Water Resources Planning and Management, 132(6), 424–433. https://doi.org/10.1061/(ASCE)0733-9496(2006)132:6(424)
- Allen, R. G., Pereira, L. S., Raes, D., & Smith, M. (1998). Crop evapotranspiration—Guidelines for computing crop water requirements. FAO Irrigation and Drainage paper 56. Rome: FAO.
- Banzhaf, H. S. (2009). Objective or multi-objective? Two historically competing visions for benefit-cost analysis. Land Economics, 85(1), 3–23. https://doi.org/10.3368/le.85.1.3
- Bauer, N., Calvin, K., Emmerling, J., Fricko, O., Fujimori, S., Hilaire, J., et al. (2017). Shared socio-economic pathways of the energy sector— Quantifying the narratives. *Global Environmental Change*, 42, 316–330. https://doi.org/10.1016/j.gloenvcha.2016.07.006
- Baum, R., Luh, J., & Bartram, J. (2013). Sanitation: A global estimate of sewerage connections without treatment and the resulting impact on MDG progress. *Environmental Science & Technology*, 47(4), 1994–2000. https://doi.org/10.1021/es304284f
 - Bekchanov, M., Sood, A., Pinto, A., & Jeuland, M. (2017). Systematic review of water-economy modeling applications. Journal of Water Resources Planning and Management, 143(8), 04017037. https://doi.org/10.1061/(ASCE)WR.1943-5452.0000793
- Berbel, J., Gutierrez-Marín, C., & Expósito, A. (2018). Microeconomic analysis of irrigation efficiency improvement in water use and water consumption. Agricultural Water Management, 203, 423–429. https://doi.org/10.1016/j.agwat.2018.02.026
- Booker, J. F., Howitt, R. E., Michelsen, A. M., & Young, R. A. (2012). Economics and the modeling of water resources and policies. *Natural Resource Modeling*, 25(1), 168–218. https://doi.org/10.1111/j.1939-7445.2011.00105.x
- Booker, J. F., Michelsen, A. M., & Ward, F. A. (2005). Economic impact of alternative policy responses to prolonged and severe drought in the Rio Grande Basin. Water Resources Research, 41, W02026. https://doi.org/10.1029/2004WR003486

- Booker, J. F., & Young, R. A. (1994). Modeling intrastate and Interstate Markets for Colorado River Water Resources. *Journal of Environmental Economics and Management*, 26(1), 66–87. https://doi.org/10.1006/jeem.1994.1005
- Briscoe, J. (1999). The financing of hydropower, irrigation and water supply infrastructure in developing countries. International Journal of Water Resources Development, 15(4), 459–491. https://doi.org/10.1080/07900629948718

Brooke, A., Kendrick, D., & Meeraus, A. (1988). GAMS: A user's guide. New York: The Scientific Press.

Brown, C., Lund, J., Cai, X., Reed, P., Zagona, E., Ostfeld, A., et al. (2015). The future of water resources systems analysis: Toward a scientific framework for sustainable water management. *Water Resources Research*, *51*, 6110–6124. https://doi.org/10.1002/2015WR017114

- Cai, X., & Rosegrant, M. (2002). Global water demand and supply projections. Water International, 27(2), 159–169. https://doi.org/10.1080/02508060208686989
- Cai, X., Rosegrant, M., & Ringler, C. (2003). Physical and economic efficiency of water use in the river basin: Implications for efficient water management. Water Resources Research, 39(1), 1013. https://doi.org/10.1029/2001WR000748
- Cai, X., Wallington, K., Shafiee-Jood, M., & Marston, L. (2018). Understanding and managing the food-energy-water nexus—Opportunities for water resources research. Advances in Water Resources, 111, 259–273. https://doi.org/10.1016/j.advwatres.2017.11.014
- Cervigni, R., Liden, R., Neumann, J. E., & Strzepek, K. M. (2015). Enhancing the climate resilience of Africa's infrastructure: The power and water sectors, Africa Development Forum series. Washington, DC: World Bank.
- Connor, J. D., Schwabe, K., King, D., Kaczan, D., & Kirby, M. (2009). Impacts of climate change on lower Murray irrigation. Australian Journal of Agricultural and Resource Economics, 53(3), 437–456. https://doi.org/10.1111/j.1467-8489.2009.00460.x

Csardi, G., & Nepusz, T. (2011). The igraph software package for complex network research. Complex Systems, 1695, 1-9.

Cuya, D. G. P., Brandimarte, L., Popescu, I., Alterach, J., & Peviani, M. (2013). A GIS-based assessment of maximum potential hydropower production in La Plata basin under global changes. *Renewable Energy*, 50, 103–114. https://doi.org/10.1016/j.renene.2012.06.019

Dalin, C., Wada, Y., Kastner, T., & Puma, M. J. (2017). Groundwater depletion embedded in international food trade. *Nature*, 543(7647), 700–704. https://doi.org/10.1038/nature21403

Draper, A., Jenkins, M., Kirby, K., Lund, J., & Howitt, R. (2003). Economic-engineering optimization for California water management. Journal of Water Resources Planning and Management, 129(3), 155–164. https://doi.org/10.1061/(ASCE)0733-9496(2003)129:3(155)

- Dubreuil, A., Assoumou, E., Bouckaert, S., Selosse, S., & Maizi, N. (2012). Water modeling in an energy optimization framework—The waterscarce middle east context. *Applied Energy*, 101, 268–279.
- Escriva-Bou, A., Lund, J. R., & Pulido-Velazquez, M. (2015). Optimal residential water conservation strategies considering related energy in California. Water Resources Research, 51, 4482–4498. https://doi.org/10.1002/2014WR016821

Escriva-Bou, A., Pulido-Velazquez, M., & Pulido-Velazquez, D. (2017). Economic value of climate change adaptation strategies for water management in Spain's Jucar Basin. Journal of Water Resources Planning and Management, 143, 1–13.

Evans, R. G., & Sadler, E. J. (2008). Methods and technologies to improve efficiency of water use. Water Resources Research, 44, W00E04. https://doi.org/10.1029/2007WR006200

Fan, Y., Li, H., & Miguez-Macho, G. (2013). Global patterns of groundwater table depth. Science, 339(6122), 940–943. https://doi.org/10.1126/ science.1229881

Food and Agriculture Organization (2016). FAO's Information System on Water and Agriculture (AQUASTAT). Rome: Food and Agriculture Organization (FAO).

Fischer, G., Nachtergaele, F.O., Prieler, S., Teixeira, E., Toth, G., van Velthuizen, H., et al. (2012). Global Agro-eccological Zones (GAEZ v3.0) model documentation. International Institute for Applied Systems Analysis.

Fischer, G., Tubiello, F. N., van Velthuizen, H., & Wiberg, D. A. (2007). Climate change impacts on irrigation water requirements: Effects of mitigation, 1990–2080. Technological Forecasting and Social Change, 74(7), 1083–1107. https://doi.org/10.1016/j.techfore.2006.05.021

Fisher, F., Arlosoroff, S., Eckstein, Z., Haddadin, M., Hamati, S. G., Huber-Lee, A., et al. (2002). Optimal water management and conflict resolution: The Middle East Water Project. *Water Resources Research*, *38*(11), 1243. https://doi.org/10.1029/2001WR000943

Fricko, O., Parkinson, S., Johnson, N., Strubegger, M., van Vliet, M., & Riahi, K. (2016). Energy sector water use implications of a 2 °C climate policy. *Environmental Research Letters*, *11*(3), 034011. https://doi.org/10.1088/1748-9326/11/3/034011

Global Administrative Areas Database, (2012). Global administrative areas. University of California Berkeley Museum of Vertebrate Zoology and the International Rice Research Institute.

Ghaffour, N., Missimer, T. M., & Amy, G. L. (2013). Technical review and evaluation of the economics of water desalination: Current and future challenges for better water supply sustainability. *Desalination*, 309, 197–207. https://doi.org/10.1016/j.desal.2012.10.015

Gohar, A. A., Ward, F. A., & Amer, S. A. (2013). Economic performance of water storage capacity expansion for food security. *Journal of Hydrology*, 484, 16–25. https://doi.org/10.1016/j.jhydrol.2013.01.005

Global Water Intelligence (2017). IDA Desalination Yearbook 2017–2018. Global Water Intelligence.

Hanasaki, N., Fujimori, S., Yamamoto, T., Yoshikawa, S., Masaki, Y., Hijioka, Y., et al. (2013). A global water scarcity assessment under shared socio-economic pathways—Part 2: Water availability and scarcity. *Hydrology and Earth System Sciences*, 17(7), 2393–2413. https://doi.org/ 10.5194/hess-17-2393-2013

- Hanasaki, N., Fujimori, S., Yamamoto, T., Yoshikawa, S., Masaki, Y., Hijioka, Y., et al. (2013). A global water scarcity assessment under shared socio-economic pathways—Part 1: Water use. *Hydrology and Earth System Sciences*, *17*(7), 2375–2391. https://doi.org/10.5194/hess-17-2375-2013
- Hanasaki, N., Yoshikawa, S., Pokhrel, Y., & Kanae, S. (2018). A global hydrological simulation to specify the sources of water used by humans. Hydrology and Earth System Sciences, 22, 789–817.

Harou, J., Pulido-Velazquez, M., Rosenberg, D., Medellín-Azuara, J., Lund, J., & Howitt, R. (2009). Hydro-economic models: Concepts, design, applications, and future prospects. *Journal of Hydrology*, 375(3-4), 627–643. https://doi.org/10.1016/j.jhydrol.2009.06.037

Hejazi, M., Edmonds, J., Clarke, L., Kyle, P., Davies, E., Chaturvedi, V., et al. (2014). Long-term global water projections using six socioeconomic scenarios in an integrated assessment modeling framework. *Technological Forecasting and Social Change*, 81, 205–226. https://doi.org/ 10.1016/j.techfore.2013.05.006

Holland, R. A., Scott, K. A., Flörke, M., Brown, G., Ewers, R. M., Farmer, E., et al. (2015). Global impacts of energy demand on the freshwater resources of nations. *Proceedings of the National Academy of Sciences*, 112, 6707–6716.

Hurford, A., Huskova, I., & Harou, J. (2014). Using many-objective trade-off analysis to help dams promote economic development, protect the poor and enhance ecological health. *Environmental Science & Policy*, *38*, 72–86. https://doi.org/10.1016/j.envsci.2013.10.003

International Energy Agency (2016). Water energy nexus, excerpt from the World Energy Outlook 2016. Paris: International Energy Agency (IEA). Iglesias, R., Ortega, E., Batanero, G., & Quintas, L. (2010). Water reuse in Spain: Data overview and costs estimation of suitable treatment trains. Desalination, 263(1-3), 1–10. https://doi.org/10.1016/j.desal.2010.06.038 Ignaciuk, A., & Mason-D'Croz, D. (2014). Modelling adaptation to climate change in agriculture. Paris: OECD Publishing. Jenkins, M. W., & Lund, J. R. (2000). Integrating yield and shortage management under multiple uncertainties. Journal of Water Resources

Planning and Management, 126(5), 288–297. https://doi.org/10.1061/(ASCE)0733-9496(2000)126:5(288) Jones, B., & O'Neill, B. C. (2016). Spatially explicit global population scenarios consistent with the Shared Socioeconomic Pathways.

Environmental Research Letters, 11(8), 084003. https://doi.org/10.1088/1748-9326/11/8/084003

Kahil, M. T., Connor, J. D., & Albiac, J. (2015). Efficient water management policies for irrigation adaptation to climate change in southern Europe. *Ecological Economics*, 120, 226–233. https://doi.org/10.1016/j.ecolecon.2015.11.004

Kahil, M. T., Dinar, A., & Albiac, J. (2015). Modeling water scarcity and droughts for policy adaptation to climate change in arid and semiarid regions. *Journal of Hydrology*, 522, 95–109. https://doi.org/10.1016/j.jhydrol.2014.12.042

Kahil, M. T., Ward, F. A., Albiac, J., Eggleston, J., & Sanz, D. (2016). Hydro-economic modeling with aquifer-river interactions to guide sustainable basin management. *Journal of Hydrology*, 539, 510–524. https://doi.org/10.1016/j.jhydrol.2016.05.057

Keller, A., Sakthivadivel, R., & Seckler, D., (2000). Water scarcity and the role of storage in development, Research report. International Water Management Institute (IWMI), Colombo.

Keppo, I., & Strubegger, M. (2010). Short term decisions for long term problems—The effect of foresight on model based energy systems analysis. Energy, 35(5), 2033–2042. https://doi.org/10.1016/j.energy.2010.01.019

Khan, Z., Linares, P., Rutten, M., Parkinson, S., Johnson, N., & García-González, J. (2017). Spatial and temporal synchronization of water and energy systems: Towards a single integrated optimization model for long-term resource planning. *Applied Energy*, 210, 499–517.

Kim, S. H., Hejazi, M., Liu, L., Calvin, K., Clarke, L., Edmonds, J., et al. (2016). Balancing global water availability and use at basin scale in an integrated assessment model. *Climatic Change*, 136(2), 217–231. https://doi.org/10.1007/s10584-016-1604-6

Kirshen, P., (2007). Adaptation options and cost in water supply. A Report for United Nations Framework Convention on Climate Change. Accessible at: http://unfccc.int/cooperation_and_support/financial_mechanism/financial_mechanism_gef/items/4054.php

Kirshen, P., McCluskey, M., Vogel, R., & Strzepek, K. (2005). Global analysis of changes in water supply yields and costs under climate change: A case study in China. Climatic Change, 68(3), 303–330. https://doi.org/10.1007/s10584-005-1148-7

Kummu, M., Ward, P. J., Moel, H., & Varis, O. (2010). Is physical water scarcity a new phenomenon? Global assessment of water shortage over the last two millennia. *Environmental Research Letters*, 5(3), 034006. https://doi.org/10.1088/1748-9326/5/3/034006

Lehner, B., & Grill, G. (2013). Global river hydrography and network routing: Baseline data and new approaches to study the world's large river systems. *Hydrological Processes*, 27(15), 2171–2186. https://doi.org/10.1002/hyp.9740

Lehner, B., Liermann, C. R., Revenga, C., Vörösmarty, C., Fekete, B., Crouzet, P., et al. (2011). High-resolution mapping of the world's reservoirs and dams for sustainable river-flow management. *Frontiers in Ecology and the Environment*, 9(9), 494–502. https://doi.org/10.1890/100125

Lele, S. M. (1987). Improved algorithms for reservoir capacity calculation incorporating storage-dependent losses and reliability norm. Water Resources Research, 23(10), 1819–1823. https://doi.org/10.1029/WR023i010p01819

Liu, J., Yang, H., Gosling, S. N., Kummu, M., Flörke, M., Pfister, S., et al. (2017). Water scarcity assessments in the past, present, and future. *Earth's Future*, 5(6), 545–559. https://doi.org/10.1002/2016EF000518

Liu, L., Parkinson, S. C., Gidden, M., Byers, E., Satoh, Y., Riahi, K., & et al. (2018). Quantifying the potential for reservoirs to secure future surface water yields in the world's largest river basins. *Environmental Research Letters*, *13*(4), 044026. https://doi.org/10.1088/1748-9326/aab2b5

Loulou, R. (2008). ETSAP-TIAM: The TIMES integrated assessment model Part II: Mathematical formulation. *Computational Management Science*, 5(1-2), 41-66. https://doi.org/10.1007/s10287-007-0045-0

Loulou, R., & Labriet, M. (2008). ETSAP-TIAM: The TIMES integrated assessment model Part I: Model structure. Computational Management Science, 5(1-2), 7-40. https://doi.org/10.1007/s10287-007-0046-z

Medellín-Azuara, J., Harou, J., Olivares, M., Madani, K., Lund, J., Howitt, R., et al. (2008). Adaptability and adaptations of California's water supply system to dry climate warming. *Climatic Change*, 87(S1), 75–90. https://doi.org/10.1007/s10584-007-9355-z

Mekonnen, M., & Hoekstra, A. (2016). Four billion people facing severe water scarcity. Science Advances, 2(2), e1500323. https://doi.org/ 10.1126/sciadv.1500323

Nasta, P., Gates, J. B., & Wada, Y. (2016). Impact of climate indicators on continental-scale potential groundwater recharge in Africa. *Hydrological Processes*, *30*(19), 3420–3433. https://doi.org/10.1002/hyp.10869

Nigatu, G., & Dinar, A. (2016). Economic and hydrological impacts of the Grand Ethiopian Renaissance Dam on the Eastern Nile River Basin. Environment and Development Economics, 21(04), 532–555. https://doi.org/10.1017/S1355770X15000352

Noel, J. E., & Howitt, R. E. (1982). Conjunctive multibasin management: An optimal control approach. Water Resources Research, 18(4), 753–763. https://doi.org/10.1029/WR018i004p00753

O'Neill, B. C., Kriegler, E., Riahi, K., Ebi, K. L., Hallegatte, S., Carter, T. R., et al. (2014). A new scenario framework for climate change research: The concept of shared socioeconomic pathways. *Climatic Change*, 122(3), 387–400. https://doi.org/10.1007/s10584-013-0905-2

Parkinson, S. C., Djilali, N., Krey, V., Fricko, O., Johnson, N., Khan, Z., et al. (2016). Impacts of groundwater constraints on Saudi Arabia's lowcarbon electricity supply strategy. Environmental Science and Technology, 50(4), 1653–1662. https://doi.org/10.1021/acs.est.5b05852

Parkinson, S. C., Johnson, N., Rao, N. D., Jones, B., van Vliet, M. T. H., Fricko, O., et al. (2016). Climate and human development impacts on municipal water demand: A spatially-explicit global modeling framework. *Environmental Modelling and Software*, 85, 266–278. https://doi. org/10.1016/j.envsoft.2016.08.002

Perry, C., & Steduto, P. (2017). Does improved irrigation technology save water? A review of the Evidence. Discussion paper on irrigation and sustainable water resources management in the Near East and North Africa. Food and Agriculture Organization of the United Nations. Cairo.

Phocaides, A. (2000). Technical handbook on pressurized irrigation techniques. Rome: FAO.

Plappally, A. K., & Lienhard, V. J. H. (2012). Energy requirements for water production, treatment, end use, reclamation, and disposal. Renewable and Sustainable Energy Reviews, 16(7), 4818–4848. https://doi.org/10.1016/j.rser.2012.05.022

Portmann, F. T., Siebert, S., & Döll, P. (2010). MIRCA2000—Global monthly irrigated and rainfed crop areas around the year 2000: A new highresolution data set for agricultural and hydrological modeling. *Global Biogeochemical Cycles*, 24, GB1011. https://doi.org/10.1029/ 2008GB003435

Qureshi, M. E., Schwabe, K., Connor, J., & Kirby, M. (2010). Environmental water incentive policy and return flows. Water Resources Research, 46, W04517. https://doi.org/10.1029/2008WR007445

Raptis, C. E., & Pfister, S. (2016). Global freshwater thermal emissions from steam-electric power plants with once-through cooling systems. *Energy*, 97, 46–57. https://doi.org/10.1016/j.energy.2015.12.107

Raskin, P., Gleick, P., Kirshen, P., Pontius, G., & Strzepek, K. (1997). Water futures: Assessment of long-range patterns and problems. Stockholm: Comprehensive Assessment of the Freshwater Resources of the World. Stockholm Environment Institute. Ringler, C., von Braun, J., & Rosegrant, M. W. (2004). Water policy analysis for the Mekong River basin. *Water International*, 29(1), 30–42. https://doi.org/10.1080/02508060408691746

- Rodriguez-Garcia, G., Molinos-Senante, M., Hospido, A., Hernández-Sancho, F., Moreira, M. T., & Feijoo, G. (2011). Environmental and economic profile of six typologies of wastewater treatment plants. *Water Research*, 45(18), 5997–6010. https://doi.org/10.1016/j. watres.2011.08.053
- Rogers, P. P., & Fiering, M. B. (1986). Use of systems analysis in water management. *Water Resources Research*, 22(9S), 146S–158S. https://doi. org/10.1029/WR022i09Sp0146S
- Rosenberg, D. E., Howitt, R. E., & Lund, J. R. (2008). Water management with water conservation, infrastructure expansions, and source variability in Jordan. *Water Resources Research*, 44, W11402. https://doi.org/10.1029/2007WR006519
- Rosenberg, D. E., Tarawneh, T., Abdel-Khaleq, R., & Lund, J. R. (2007). Modeling integrated water user decisions in intermittent supply systems. Water Resources Research, 43, W07425. https://doi.org/10.1029/2006WR005340
- Sanchez, B., Iglesias, A., McVittie, A., Alvaro-Fuentes, J., Ingram, J., Mills, J., et al. (2016). Management of agricultural soils for greenhouse gas mitigation: Learning from a case study in NE Spain. *Journal of Environmental Management*, 170, 37–49. https://doi.org/10.1016/j. jenvman.2016.01.003
- Satoh, Y., Kahil, T., Byers, E., Burek, P., Fischer, G., Tramberend, S., et al. (2017). Multi-model and multi-scenario assessments of Asian water futures: The water futures and solutions (WFaS) initiative. *Earth's Future*, *5*(7), 823–852. https://doi.org/10.1002/2016EF000503
- Sauer, T., Havlík, P., Schneider, U. A., Schmid, E., Kindermann, G., & Obersteiner, M. (2010). Agriculture and resource availability in a changing world: The role of irrigation. *Water Resources Research*, *46*, W06503. https://doi.org/10.1029/2009WR007729
- Schmidt-Traub, G. (2015). Investment needs to achieve the sustainable development goals, SDSN Working Paper. Sustainable Development Solution Netwrok (SDSN)
- Stillwell, A. S., King, C. W., & Webber, M. E. (2010). Desalination and long-haul water transfer as a water supply for Dallas, Texas: A case study of the energy-water Nexus in Texas. *Texas Water Journal*, *1*, 9.
- Strzepek, K., Schlosser, A., Gueneau, A., Gao, X., Blanc, E., Fant, C., et al. (2013). Modeling water resource systems within the framework of the MIT Integrated Global System Model: IGSM-WRS. Journal of Advances in Modeling Earth Systems, 5, 638–653. https://doi.org/10.1002/ jame.20044
- Taliotis, C., Shivakumar, A., Ramos, E., Howells, M., Mentis, D., Sridharan, V., et al. (2016). An indicative analysis of investment opportunities in the African electricity supply sector—Using TEMBA (The Electricity Model Base for Africa). *Energy for Sustainable Development*, *31*, 50–66. https://doi.org/10.1016/j.esd.2015.12.001
- Tanaka, S. K., Zhu, T., Lund, J. R., Howitt, R. E., Jenkins, M. W., Pulido, M. A., et al. (2006). Climate warming and water management adaptation for California. *Climatic Change*, 76(3-4), 361–387. https://doi.org/10.1007/s10584-006-9079-5
- United Nations (2017). 2015 Energy Statistics Yearbook. Department of Economic and Social Affairs. New York: Statistics Division. United Nations.
- van Beek, L. P. H., Wada, Y., & Bierkens, M. F. P. (2011). Global monthly water stress: 1. Water balance and water availability. *Water Resources Research*, 47, W07517. https://doi.org/10.1029/2010WR009791
- Veldkamp, T. I. E., Wada, Y., Aerts, J. C. J. H., Döll, P., Gosling, S. N., Liu, J., et al. (2017). Water scarcity hotspots travel downstream due to human interventions in the 20th and 21st century. *Nature Communications*, 8. https://doi.org/10.1038/ncomms15697
- Wada, Y., Bierkens, M.F.P., De Roo, A., Dirmeyer, P.A., Famiglietti, J.S., Hanasaki, N., et al. (2017). Human-water interface in hydrological modelling: Current status and future directions. *Hydrology and Earth System Sciences*, 21(8), 4169–4193. https://doi.org/10.5194/hess-21-4169-2017
- Wada, Y., Flörke, M., Hanasaki, N., Eisner, S., Fischer, G., Tramberend, S., et al. (2016). Modeling global water use for the 21st century: The Water Futures and Solutions (WFaS) initiative and its approaches. *Geoscientific Model Development*, 9(1), 175–222. https://doi.org/10.5194/gmd-9-175-2016
- Wada, Y., Gleeson, T., & Esnault, L. (2014). Wedge approach to water stress. Nature Geoscience, 7(9), 615–617. https://doi.org/10.1038/ ngeo2241
- Wada, Y., van Beek, L. P. H., van Kempen, C. M., Reckman, J. W. T. M., Vasak, S., & Bierkens, M. F. P. (2010). Global depletion of groundwater resources. *Geophysical Research Letters*, 37, L20402. https://doi.org/10.1029/2010GL044571
- Wada, Y., van Beek, L. P. H., Viviroli, D., Dürr, H. H., Weingartner, R., & Bierkens, M. F. P. (2011). Global monthly water stress: 2. Water demand and severity of water stress. *Water Resources Research*, 47, W07518. https://doi.org/10.1029/2010WR009792
- Wada, Y., van Beek, L. P. H., Wanders, N., & Bierkens, M. F. P. (2013). Human water consumption intensifies hydrological drought worldwide. *Environmental Research Letters*, 8(3), 034036. https://doi.org/10.1088/1748-9326/8/3/034036
- Wada, Y., Wisser, D., & Bierkens, M. F. P. (2014). Global modeling of withdrawal, allocation and consumptive use of surface water and groundwater resources. *Earth System Dynamics*, 5(1), 15–40. https://doi.org/10.5194/esd-5-15-2014
- Ward, F. A. (2009). Economics in integrated water management. Environmental Modelling & Software, 24(8), 948–958. https://doi.org/10.1016/j.envsoft.2009.02.002
- Ward, F. A., & Lynch, T. P. (1997). Is dominant use management compatible with basin-wide economic efficiency? Water Resources Research, 33(5), 1165–1170. https://doi.org/10.1029/97WR00273
- Ward, F. A., & Pulido-Velazquez, M. (2008). Water conservation in irrigation can increase water use. Proceedings of the National Academy of Sciences of the United States of America, 105(47), 18,215–18,220. https://doi.org/10.1073/pnas.0805554105
- Ward, P. J., Pauw, P., Brander, L. M., Aerts, J. C. J. H., & Strzepek, K. (2010). Costs of adaptation related to industrial and municipal water supply and riverine flood protection, Discussion Paper. Washington, DC: World Bank.
- Wiberg, D., & Strzepek, K. (2005). Development of regional economic supply curves for surface water resources and climate change assessments: A case study of China, Research Report. International Institute for Applied Systems Analysis (IIASA), Laxenburg.
- Wisser, D., Fekete, B. M., Vörösmarty, C. J., & Schumann, A. H. (2010). Reconstructing 20th century global hydrography: A contribution to the global terrestrial network- hydrology (GTN-H). *Hydrology and Earth System Sciences*, 14(1), 1–24. https://doi.org/10.5194/hess-14-1-2010 Woetzel, J., Garemo, N., Mischke, J., Hjerpe, M., & Palter, R., (2016). Bridging global infrastructure gaps. McKinsey Global Institute.
- Zawahri, N., Dinar, A., & Nigatu, G. (2016). Governing international freshwater resources: An analysis of treaty design. *International Environmental Agreements: Politics, Law and Economics*, *16*(2), 307–331. https://doi.org/10.1007/s10784-014-9259-0